Author’s response to reviews

Title: Size and Position of the Healthy Meniscus, and its Correlation with Sex, Height, Weight, Bone Size, and Age - a cross-sectional study

Authors:

Katja Bloecker (katja.bloecker@pmu.ac.at)
Wolfgang Wirth (wolfgang.wirth@pmu.ac.at)
Martin Hudelmaier (martin.hudelmaier@pmu.ac.at)
Rainer Burgkart (burgkart@ortho.med.tum.de)
Richard Frobell (richard.frobell@med.lu.se)
Felix Eckstein (felix.eckstein@pmu.ac.at)

Version: 2 Date: 25 March 2011

Author’s response to reviews:

Thank you very much for your suggestions. I changed the format of the abstract, adding the expected headlines.

Additionally I added a paragraph in the methods section containing the approval nr and the institution that approved the study to state that this study conformed to the Helsinki Declaration (page 5, line 6-8)